Author's response to reviews

Title: Individuals with chronic low back pain have greater difficulty in engaging in positive lifestyle behaviours than those without back pain: An assessment of health literacy.

Authors:
Andrew M Briggs (A.Briggs@curtin.edu.au)
Joanne E Jordan (joannejordan14@gmail.com)
Peter B O’Sullivan (P.OSullivan@curtin.edu.au)
Rachelle Buchbinder (Rachelle.Buchbinder@monash.edu.au)
Angus F Burnett (a.burnett@cuhk.edu.hk)
Richard H Osborne (Richard.Osborne@deakin.edu.au)
Leon M Straker (L.Straker@curtin.edu.au)

Version: 3 Date: 2 July 2011

Author's response to reviews:

Dear Abigail
Thank you for your email dated 1 July 2011. As requested, I have changed the term "Introduction" to "Backgound" in the paper so that it is consistent with the Journal's instructions. I hope this change is satisfactory. The rest of the paper seems to align with the Journal's instructions.
I look forward to hearing from you,

Sincerely

Andrew